Author's response to reviews

Title: Tracheo-Brachiocephalic Artery Fistula After Tracheostomy Associated With Thoracic Deformity: A Case Report

Authors:

Kie Ogawa (kieo@belle.shiga-med.ac.jp)
Norihisa Nitta (r34nitta@belle.shiga-med.ac.jp)
Akinaga Sonoda (akinagasonoda@yahoo.co.jp)
Masashi Takahashi (masashi@belle.shiga-med.ac.jp)
Tomoaki Suzuki (suzuki@belle.shiga-med.ac.jp)
Noriaki Tezuka (tezuka@belle.shiga-med.ac.jp)
Jun Hanaoka (hanaoka@belle.shiga-med.ac.jp)
Shoji Kitamura (shoji@belle.shiga-med.ac.jp)
Kiyoshi Murata (murata@belle.shiga-med.ac.jp)

Version: 2 Date: 20 July 2011

Author's response to reviews:

Journal of Medical Case Reports
Editor-in-Chief
Michael Kidd AM
Dear Dr Kidd

I am sorry to sending you our incorrect manuscript. I modified our manuscript following author instruction form. Please let me know if there is anything we should edit more.

Sincerely, Yours

Norihisa Nitta (corresponding author)
Department of Radiology
Shiga University of Medical Science